Author’s response to reviews

Title: A repeated measurement study investigating the impact of school outdoor environment upon physical activity across ages and seasons in Swedish second, fifth and eighth graders

Authors:

Peter Pagels (peter.pagels@lnu.se)
Anders Raustorp (anders.raustorp@ped.gu.se)
Antonio Ponce De Leon (Antonio.Ponce.De.Leon@ki.se)
Fredrika Mårtensson (fredrika.martensson@slu.se)
Maria Kylin (maria.kylin@slu.se)
Cecilia Boldemann (cecilia.boldemann@ki.se)

Version: 4 Date: 29 July 2014

Author’s response to reviews:

Editorial comments:

"Please add a formal parental consent statement to the manuscript, stating whether consent was written or verbal (for parents and children)."

Response to Editorial comments:

We made a point of that in the methods section (lines 117 - 118): “Both parents and their participating children signed a written consent to participate in the study.”

We hope that this is ok.

Sincerely,

Peter Pagels